Author's response to reviews

Title: Antihypertensive potential of the aqueous extract which combine leaf of Persea americana Mill. (Lauraceae), stems and leaf of Cymbopogon citratus (D.C) Stapf. (Poaceae), fruits of Citrus medical L. (Rutaceae) as well as honey in ethanol and sucrose experimental of model

Authors:

Paul Désiré D Dzeufiet (dzeufiet@yahoo.fr)
Amelie Moguéo (amelie.mogue@yahoo.fr)
Danielle Claude Bilanda (bilandad@yahoo.com)
Bibi-Farouk O Aboubakar (bibifarouck@yahoo.fr)
Léonard Tedong (ltedong@yahoo.fr)
Théophile Dimo (dimo59@yahoo.com)
Pierre Kamtchouing (pikam55@yahoo.fr)

Version: 4
Date: 29 October 2014

Author's response to reviews: see over
Dear Editor

Please find below the reply to the reviewer report concerning our manuscript entitled “Antihypertensive potential of the aqueous extract which combine leaf of *Persea americana* Mill. (Lauraceae), stems and leaf of *Cymbopogon citratus* (D.C) Stapf. (Poaceae), fruits of *Citrus medical* L. (Rutaceae) as well as honey in ethanol and sucrose experimental of model.”

We did the necessary correction in the manuscript as well as figures and tables.

Best regards.

---

**Reply to reviewer**

We are grateful to the reviewer for his comments and suggestions. The corrections are highlighted in the manuscript, tables and figures.

**Remark 1:** Check English throughout the MS, consistency using abbreviation.

**Answer:** We have check English and abbreviations accordingly

**Remark 2:** Changing alcohol to ethanol

Thickness of the aorta, SE and n should be shown.

**Answer:** This has been done in the manuscript, figures and tables as suggested.

We added the standard error ±15.31 and the number of measurement which was previously included in the methodology ([Six arteries per animal were quantified, given a total of 36 vessels per group](#)).

**Remark 3:** Figure 7 needs standard bar.

**Answer:** Standard bar has been included in the figure.

**References:** Abbreviation of the journal.

**Answer:** We have check abbreviation of the journal in references.